Author’s response to reviews

Title: Pattern recognition in menstrual bleeding diaries by statistical cluster analysis

Authors:

Christoph Gerlinger (christoph.gerlinger@bayerhealthcare.com)
Jens Wessel (jens.wessel@bayerhealthcare.com)
Gerd Kallischnigg (gerd.kallischnigg@tu-berlin.de)
Jan Endrikat (jan.endrikat.b@bayer.com)

Version: 4 Date: 8 July 2009

Author’s response to reviews:

Dear Editor,

all formatting changes requested have been performed on the manuscript. However, we would like you to perform the figure cropping.

Thank you!

Kind regards,

Christoph Gerlinger